

MANUSCRIPT NUMBER 21327

ANSWERS TO REVIEWERS

Dear Editor, Authors Thank you for sending the manuscript "ADHESIVE SMALL BOWEL ADHESIONS OBSTRUCTION [ASBO]: evolutions in diagnosis, management and prevention" - Post operative adhesions are common significant problem, but most of presented informations do not reflect novel idea - Paper can be rewritten with stress on preventive measures - Frequent grammatical and spelling mistakes Thanks

Dear Reviewer,

unfortunately, there are not yet devices able to totally prevent the intraperitoneal adhesions formation after abdominal surgery; only the use of correct surgical technique and the avoidance of traumatic intraperitoneal organs maneuvers may help to reduce postoperative adhesions incidence.